\documentstyle[12pt,twoside]{article}
\input{macros-course}
\long\def\ignore#1{}
\begin{document}

\handout{final-info}{Final Examination Information}{November 25, 1997}

\vspace*{.2in}

\noindent \underline{WHEN:} {\large Thursday, December 18, 9:00 am  to 12:00 noon,
 in Dupont}\\

\vspace*{.2in}

%\underline{Students with conflicts only}: {\large Wednesday, December 18, 9:00 AM, in 26-204}\\

\vspace*{.2in}

\noindent \underline{CRIB SHEET:} {\large The final is not open-book, but you are 
allowed to bring two 2-sided sheets to the exam. You may have whatever 
information you 
like on it but it must be handwritten by you and must not be xeroxed.}

\vspace*{.2in}

\noindent \underline{NO CALCULATORS ALLOWED}

\vspace*{.2in}

\noindent \underline{SYLLABUS:} {\large Everything done until and including the 
Lect. 26 on December 9.}

\vspace*{.2in}

%\underline{Make-up Exam:} {\large If you have a conflict come see us NOW.}

%\vspace*{.2in}

\noindent \underline{SPECIAL OFFICE HOURS:} {\large There will be special office hours 
for the week of December 1}:

\( \begin{array}{llll}
Russell: & Wednesday & December~2, & 5-8pm\\
Alex: & Thursday, & December~4, & 11:30am-2:30pm\\
David: & Thursday, & December~4, & 5-8pm \mbox{ in } 2-342\\    
Santosh: & Tuesday, & December~2, & 2-4pm\\
Gunnar: & Thursday, & December~4, & 11am-2pm\\
\end{array} \)
\vspace*{.3in}

\noindent {\large There will also be special office hours on the week of the final, that will be
announced next week}

%\underline{Special office hours:} {\large There will be special office hours 
%during final-week as follows}:

%\( \begin{array}{llll}
%Russell: & Wednesday & December~17, & 1-4pm\\
%Alex: & Monday and Wednesday, & December~15,17, & 2-5\\
%David: & ? & December~?, & ?\\    
%Santosh: & Tuesday & December~16, & 2-4pm\\
%Gunnar: & ? & December~?, & ?\\
%\end{array} \)
%\vspace*{.3in}

\end{document}

